Author’s response to reviews

Title: A 9-year-old Korean girl with Fontaine progeroid syndrome: a case report with further phenotypical delineation and description of clinical course during long-term follow-up

Authors:

Jaehui Ryu (jhryu25@snu.ac.kr)
Jung Min Ko (jmko@snu.ac.kr)
Choong-Ho Shin (chshinpd@snu.ac.kr)

Version: 3 Date: 08 Nov 2019

Author’s response to reviews:

1. Availability of data and materials

a) Although you have mentioned that you deposited your data in the SNUCH-PEGD repository in the Availability of data and materials section of your manuscript, you have not provided the relevant accession numbers. Please provide the relevant accession numbers or the appropriate web links corresponding to your data sets.

Please ensure that all data is already released and that all accession numbers/web links are in the ‘Availability of Data and Materials’ in their final form, and that all accession numbers/web links allow access to public data.

b) Please also provide the full name of the repository in which your data is stored in the Availability of data and materials section of your manuscript.

We finally thought and decided that our datasets generated during and/or analyzed during the current study are not publicly available because it is possible that individual privacy could be compromised. However, datasets are available from the corresponding author on reasonable request through personal contact.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

We check the whole files to be uploaded.

Thank you for your kind consideration.

Best regards,

Jung Min Ko